Author’s response to reviews

Title: Screening for infectious diseases of asylum seekers upon arrival: the necessity of the moral principle of reciprocity

Authors:

Dorien Tecla Beeres (d.t.beeres@umcg.nl)
Darren Cornisch (daznl@hotmail.com)
Machiel Vonk (machiel.vonk@ggd.groningen.nl)
Sofanne Ravensbergen (s.j.ravensbergen@umcg.nl)
Els Maeckelberghe (e.l.m.maeckelberghe@umcg.nl)
Pieter Boele van Hensbroek (p.boele@rug.nl)
Ymkje Stienstra (y.stienstra@umcg.nl)

Version: 2 Date: 20 Jan 2018

Author’s response to reviews:

BMC Medical Ethics
Gry Wester
https://bmcmedethics.biomedcentral.com/

January 20th, 2018

Subject: Submission of the revised paper II (METH-D-17-00112)

Dear Gry Wester,

Thank you for the email and the decision that has been made. The authors are very pleased to hear that the manuscript is accepted for publication in BMC Medical Ethics.
The requested change has been made.

Comment: Please remove the heading ‘Main text’ on p 3, line 91, and replace with a different heading (or no heading).

Reply: Pg 3, line 91 - The title ‘main text’ has been removed and replaced by the subheading:

“1. The principle of reciprocity as a mid-level principle for infectious disease control”

Yours sincerely,

Dorien Beeres

Corresponding author

Email: d.t.beeres@umcg.nl